Author’s response to reviews

Title: Long-duration General Anesthesia Influences the Intelligence of School Age Children

Authors:
Qingqing Zhang (pellt@163.com)
Yuanzhi Peng (baonba1982@yahoo.com.cn)
Yingwei Wang (wangyw032417@163.com)

Version: 2 Date: 07 Dec 2017

Author’s response to reviews:

Lingling Tian
BMC Anesthesiology

Re: BANE-D-17-00222R1, ”Long-duration General Anesthesia Influences the Intelligence of School Age Children”.

Dear Lingling Tian,

We have added the requested changes to the manuscript:


2. E-mail addresses for all the authors on the title page

3. Abstract includes Background, Methods, Results and Conclusions but not Aim now.

4. Section headings are 'Introduction' and 'Methods' now

5. A reference number for the ethics approval in 'Ethics approval and consent to participate' section has been added

6. The role of the author with initials YW in Authors' Contributions is “drafted the manuscript and approved the final version” now.

Sincerely Yours,
Yingwei Wang

Department of Anesthesiology, Huashan Hospital Affiliated to Fudan University, 12 Wulumuqi Middle Road, Shanghai 200040, China.
Tel: +86-13918690528
Email: wangyw032417@163.com